Author's response to reviews

Title: The Role of PET/CT for the Detection of Gastric Cancer Recurrence

Authors:

Sung Hoon Sim (boron@hanmail.net)
Yu Jung Kim (cong1005@medimail.co.kr)
Do-Youn Oh (ohdoyoun@snu.ac.kr)
Se-Hoon Lee (shlee119@snu.ac.kr)
Dong-Wan Kim (dwkimmd@chol.com)
Won Jun Kang (mdkwj@snu.ac.kr)
Seock-Ah Im (moisa@snu.ac.kr)
Tae-You Kim (kimty@snu.ac.kr)
Woo Ho Kim (woohokim@snu.ac.kr)
Dae Seog Heo (heo1013@plaza.snu.ac.kr)
Yung-Jue Bang (bangyj@plaza.snu.ac.kr)

Version: 7 Date: 27 February 2009

Author's response to reviews:

Dear Editor-in-chief

We have revised the manuscript as below:

1. We have deleted the space in the second sentence in the second paragraph in page 9. (bold)

When the image findings between the two methods were discordant, treatment decision was made according to the PET/CT findings in 7 out of 13 cases (Figure 1 and Figure 2).

2. We have corrected the alignment of tables.

Thank you.

Sincerely yours,

Do youn, Oh